Author’s response to reviews

Title: First and second trimester urinary metabolic profiles and fetal growth restriction: an exploratory nested case-control study within The Infant Development and Environment Study

Authors:

Gauri Luthra (gauri.luthra@gmail.com)
Ivan Vuckovic (vuckovic.ivan@mayo.edu)
Ananta Bangdiwala (ananta@umn.edu)
Heather Gray (heather2gray@yahoo.com)
L Bruce Redmon (redmo001@umn.edu)
Emily Barrett (Emily.barrett@eohsi.rutgers.edu)
Sheela Sathyanarayana (sheela.sathyanarayana@seattlechildrens.org)
Ruby Nguyen (nguyen@epi.umn.edu)
Shanna Swan (shanna.swan@mssm.edu)
Song Zhang (zhang.song@mayo.edu)
Petras Dzeja (dzeja.petras@mayo.edu)
Slobodan Macura (macura@mayo.edu)
K Sreekumaran Nair (nair.k@mayo.edu)

Version: 2 Date: 15 Jan 2018

Author’s response to reviews:

January 14, 2018

Janelle Coore

BMC Pregnancy and Childbirth
Dear Janelle Coore:

We are very grateful to your efforts and useful comments. We have revised the manuscript to reflect them.

We thank you for the opportunity to resubmit our manuscript. Please let me know if you have any additional questions or comments.

Sincerely,

Gauri Luthra, MD
Maternal Fetal Medicine
University of Minnesota
gauri.luthra@gmail.com